Author's response to reviews

Title: Effect of Nigella sativa fixed and essential oils on antioxidant status, hepatic enzymes, and immunity in streptozotocin induced diabetes mellitus

Authors:

Muhammad Tauseef Sultan (tauseefsultan@hotmail.com)
Masood Sadiq Butt (drmsbutt@yahoo.com)
Roselina Karim (rosaz@putra.upm.edu.my)
Shahzad Zafar Iqbal (shahzad10542005@yahoo.com)
Shakeel Ahmad Ahmad (shakeel.agronomy@gmail.com)
Muhammad Zia-Ul-Haq (ahirzia@gmail.com)
Luigi Aliberti (luigialiberti83@libero.it)
Vincenzo De Feo (defeo@unisa.it)

Version: 4 Date: 15 April 2014

Author's response to reviews: see over
Edi tor
BM C Com pl em entary and A l ternati veM edicine

Subject: Submission of revised manuscript entitled “Effect of Nigella sativa fixed and essential oils on antioxidant status, hepatic enzymes, and immunity in streptozotocin induced diabetes mellitus” for publication consideration in BMC Complementary and Alternative Medicine

Dear Editor

I have incorporated all the points raised by the reviewers and revised the manuscript accordingly (responses to the reviewers are attached as Appendix-I). I am sure that now paper is according to “BMC Complementary and Alternative Medicine”. The reviewer and editorial board comments really helped in improving the article significantly. It will be highly appreciated if the acceptance of the article may kindly be granted.

Regards

* Corresponding Author:

Prof. Vincenzo De Feo

Department of Pharmacy,
University of Salerno
Via Giovanni Paolo II, 132,
84084 Fisciano, Salerno, Italy
Phone +39 089 969751, fax +39 089 969602
<table>
<thead>
<tr>
<th>Sr. #</th>
<th>Comments</th>
<th>Response to the comments</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>It is better to add this article to the background/introductory part:</td>
<td>The article has been added in the background/introduction section and reference database has been updated</td>
</tr>
<tr>
<td>2</td>
<td>Please also ensure that your revised manuscript conforms to the journal style (<a href="http://www.biomedcentral.com/info/ifora/medicine_journals">http://www.biomedcentral.com/info/ifora/medicine_journals</a>). It is important that your files are correctly formatted.</td>
<td>The article has been formatted according to the journal format.</td>
</tr>
</tbody>
</table>